%\subsection{Publications and Presentations}
%\begin{tabular}{lr}
%Refereed publications (in print or accepted): & 17 \\
%Invited talks: & 7 \\
%All talks and poster presntations: & 21 \\
%\end{tabular}

\renewcommand{\bibname}{Related and Significant Publications}
%\nocite{Snyder:1986:CDM}
%\nocite{Bernholdt:1986:JCP-85-5120}
%\nocite{Bernholdt:1987:THEOCHEM-153-175}
%\nocite{Bernholdt:1988:JCP-89-3612}
%\nocite{Gordon:1989:CPL-158-189}
%\nocite{Augspurger:1990:JCC-11-972}
%\nocite{Stanton:1990:JCC-11-58}
%\nocite{Bartlett:1993:CCS}
%\nocite{Perera:1994:IJQC-49-559}
\nocite{Harrison:1995:JCC-17-124}
%\nocite{Guest:1994:GCC}
\nocite{Bernholdt:1995:JCP-102-9582}
%\nocite{Guest:1995:HPC}
\nocite{Bernholdt:1995:IJQCQCS-29-475}
\nocite{Guest:1996:APD}
\nocite{Bernholdt:1996:CPL-250-447}
\nocite{Feller:1996:JCP-105-1940}
\nocite{Guest:1996:FGCS-12-273}
%\nocite{Bernholdt:1997:TRAC-16-230}
\nocite{Kandaswamy:inpress:ESA}
%\nocite{Kandaswamy:inpress:OEH}
\nocite{Bernholdt:submitted:FBS}
\nocite{nwchem3.1}

\bibliographystyle{plain}
%\bibliographystyle{cv}
%\bibliography{deb-bio-refs}
\bibliography{bernholdt,nwchem}
